Author's response to reviews

Title: Reporting Recommendations for Tumor Marker Prognostic Studies
(REMARK): Explanation and Elaboration

Authors:

Douglas G Altman (doug.altman@csm.ox.ac.uk)
Lisa M McShane (McShaneL@CTEP.NCI.NIH.gov)
Willi Sauerbrei (wfs@imbi.uni-freiburg.de)
Sheila E Taube (setaube@aol.com)

Version: 3 Date: 22 December 2011

Author's response to reviews:

The only change from the version we submitted in August is that er replaced figure 4 by a different figure and the citation (ref 195) is of course different to correspond.